Author's response to reviews

Title: How Effective is Collagen Resorbable Membrane Placement after Partially Impacted Mandibular Third Molar Surgery on Postoperative Morbidity: A Prospective Randomized Comparative Study

Authors:

Adnan Kılınç (adnankilin@yahoo.com)

Mert Ataol (ataolmert@gmail.com)

Version: 2 Date: 08 Aug 2017

Author’s response to reviews:

Dear Editor;

We thank you for your interest for our manuscript with the number of OHEA-D-17-00129R1 with the title of "How Effective is Collagen Resorbable Membrane Placement after Partially Impacted Mandibular Third Molar Surgery on Postoperative Morbidity: A Prospective Randomized Comparative Study".

We have completed all the recommended corrections.

Our responses to your comments are explained below.

Best Wishes,

Dr. Adnan Kılınç

Editor Comments and responses

1) The editor suggested that we should include the email addresses of all co-authors on the title page.

We added the email addresses of all co-authors on the title page.

2) The editor suggested that we should remove the sentence "This study was supported by the Scientific Research Project Fund of Ataturk University (BAP 2016/091)" on the title page as it is already in the Declarations section.
This correction was made.

3) The editor suggested that we should remove the sentence "The summary of this study will be presented as a oral presentation in Turkish Association of Oral and Maxillofacial Surgery 24th International Scientific Congress, 23-27 May 2017, Turkey." on the title page.

This correction was made.

4) The editor suggested that we should include the Abstract as part of the submission.

We added the abstract to the main manuscript.

5) The editor suggested that we should include the heading Keywords, with a list of three to ten keywords representing the main content of the article.

This correction was made.

6) The editor suggested that we should move the List of abbreviations to after the Conclusions, in the main manuscript.

This correction was made.

7) The editor suggested that we should include the heading "Declarations", after the List of Abbreviations.

This correction was made.

8) The editor suggested that we should confirm that consent for them to participate in the study was obtained by a parent or legal guardian, if any of the patients were younger than 16yo.

We revised “Ethics approval and consent to participate” part as follows;

“… and informed consent was obtained from all patients or parents or legal guardians.”

9) The editor suggested that we should confirm if the consent was written or verbal. If verbal, please state the reason, and whether the ethics committee approved this procedure.

We revised “Ethics approval and consent to participate” part as follows;
“Informed consent to participate was obtained in writing from participants …”